Author’s response to reviews

Title: Urban Health in Africa: A critical global public health priority

Authors:

Jo Vearey (jovearey@gmail.com)

Isaac Luginaah (iluginaa@uwo.ca)

Ng"weina Magitta (ngweina.magitta@gmail.com)

Dativa Shilla (sopdatty@yahoo.com)

Tolullah Oni (tolullah.oni@mrc-epid.cam.ac.uk;Tolu.Oni@mrc-epid.cam.ac.uk)

Version: 2 Date: 22 Feb 2019

Author’s response to reviews:

Thank you for your comments.

As advised, we are re-submitting the manuscript for consideration as a "Correspondence" manuscript type. Accordingly, we note that the only suggestion from reviewer 1 to insert tables falls outside the format for this manuscript type.

The original article was already formatted to the "Correspondence" format, and so is re-submitted here.

Thank you for your consideration.

Kind Regards

Prof Tolu Oni